Author's response to reviews

Title: Designing a strategy to implement cost-effective blood transfusion management in elective hip and knee arthroplasties: a study protocol

Authors:

Veronique M.A. Voorn (v.m.a.voorn@lumc.nl)
Perla J. Marang-van de Mheen (p.j.marang@lumc.nl)
Cynthia So-Osman (c.so@sanquin.nl)
Thea P.M. Vliet Vlieland (t.p.m.vliet_vlieland@lumc.nl)
Ankie W.M.M. Koopman-van Gemert (a.w.m.m.gemert@asz.nl)
Rob G.H.H. Nelissen (r.g.h.h.nelissen@lumc.nl)
Leti van Bodegom-Vos (l.vanbodegom-vos@lumc.nl)

Version: 2 Date: 7 May 2012

Author's response to reviews:

Dear Sir,

We hereby submit the revised manuscript MS: 1085794441718832 including an outline of the timings for the research in the method section (page 6 lines 8-15).

We look forward to hearing your decision.

Regards,

Leti van Bodegom-Vos
(for the authors)